Research Paper



Immediate Effects of Kinesio and Dynamic Tapes on Postural Stability and Time to Stabilization in Women With Patellofemoral Pain Syndrome

Ali Yalfani^{1*} 💿, Shirin Rasouli¹ 💿, Zahra Raeisi² 💿

1. Department of Exercise Rehabilitation, Faculty of Sport Sciences, Bu-Ali Sina University, Hamedan, Iran.

2. Department of Sports Injuries and Corrective Exercises, Faculty of Sport Sciences, Arak University, Arak, Iran.



Citation Yalfani A, Rasouli Sh, Raeisi Z. Immediate Effects of Kinesio and Dynamic Tapes on Postural Stability and Time to Stabilization in Women With Patellofemoral Pain Syndrome. Physical Treatments. 2025; 15(1):41-50. http://dx.doi. org/10.32598/ptj.15.1.628.1

doj http://dx.doi.org/10.32598/ptj.15.1.628.1

Article info:

Received: 13 Jun 2024 Accepted: 08 Sep 2024 Available Online: 01 Jan 2025

ABSTRACT

Purpose: Patellofemoral pain syndrome (PFPS) is a common condition characterized by pain in the front of the knee. Patellar taping can be an effective intervention for reducing painful symptoms.

Methods: A total of 39 women with PFPS (aged 20-30 years) participated in this randomized crossover trial. The CoP oscillations during the single-leg stance test and TTS after descending the step were evaluated and recorded on three separate days under the following conditions: without tape (WT or control), with DT and with KT. This was done using a foot pressure measurement system. The data were analyzed using repeated measures analysis of variance with a significance level of P \leq 0.05.

Results: Comparing three different conditions demonstrated that the CoP sway, including the mediolateral and anteroposterior oscillations, sway area, path length, and average velocity of oscillations significantly decreased in the eyes-open and eyes-closed states during the DT condition as compared to the WT (P<0.05). Furthermore, the TTS significantly decreased in the DT condition compared to the WT (P<0.001) and KT (P=0.007) conditions. KT reduced CoP variables only in the EsC condition (P<0.05).

Conclusion: DT was more effective than KT and this intervention enhances postural control in women with PFPS. Given its positive effects, it is recommended that future research investigate the long-term impact of utilizing this intervention.

Keywords:

Anterior knee pain syndrome, Posture balance, Tape, Knee joint

* Corresponding Author: Ali Yalfani, Professor: Address: Department of Exercise Rehabilitation, Faculty of Sport Sciences, Bu-Ali Sina University, Hamedan, Iran. Phone: +98 (81) 38303241 E-mail: yalfani@basu.ac.ir



Copyright © 2025 The Author(s);

This is an open access article distributed under the terms of the Creative Commons Attribution License (CC-By-NC: https://creativecommons.org/licenses/by-nc/4.0/legalcode.en), which permits use, distribution, and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Highlights

• Dynamic tape (DT) had a better immediate effect than kinesio tape (KT) on postural sway and time to stabilization (TTS) in women with patellofemoral pain syndrome (PFPS).

- KT was effective on postural fluctuations in women with PFPS.
- KT failed to shorten the TTS in women with PFPS.

Plain Language Summary

As the patellofemoral joint is the least stable joint in the lower limbs, it experiences significant force (up to several times the body weight) during a wide range of movements involved in activities of daily living. PFPS is the most common knee disorder among young adults. It is often referred to as a black hole due to its unexplained nature, with no known surgical or therapeutic interventions to solve this dysfunction. Women have more knee joint laxity compared to men. This is also associated with lower proprioception in their joints, which can result in less stability, a higher risk of ligament and soft tissue damage, and loss of balance in women compared to men. One of the commonly used conservative treatments for PFPS is taping, which is considered to be more effective due to its various benefits. Kinesio tape is commonly used now; however, dynamic tape is relatively new and still less known. In this study, we compared the immediate effect of these two tapes, and the results showed that the dynamic tape was more useful for women with PFPS. Therefore, we suggest the immediate use of this type for women with PFPS and also recommend checking the long-term effects of using dynamic tape.

Introduction



atellofemoral pain syndrome (PFPS) refers to the pain in the patella (kneecap) or its surroundings induced by physical and biomechanical changes in the patellofemoral joint (PFJ), with an incidence

rate of 15% to 25% [1]. Multiple causes have been documented for this condition to date, including patellar malalignment, joint stiffness, muscle imbalance around the knee extensors and joint overuse [2]. Of note, PFPS may lead to various complications, including impaired balance [3], which can result in an increased risk of slips and falls during activities of daily living, such as walking [4]. Maintaining balance is crucial for performing everyday tasks and sports activities independently, effectively, and without relying on others [5, 6]. On the other hand, controlling posture and maintaining static and dynamic balance are influenced by visual, vestibular, and proprioceptive systems [7]. Given that proprioception and balance are related to each other in adults [8], the former has been identified as the primary source in the sensory system for maintaining balance. Proprioception is also the leading afferent component of the motor system and an essential one in the motion control system. It significantly contributes to joint dynamics and prevents chronic injuries and degenerative joint diseases [9, 10].

As evidenced, impaired proprioception is typically seen in individuals with PFPS compared to healthy cases [11]. There is also a disorder in the proprioception of the knee joint on the affected side and a decrease in this regard on the unaffected side [12]. In addition, some changes in the movement patterns, such as walking velocity and reduced knee flexion angle in the static phase of the gait, have been reported so far [13]. Considering this, proprioception plays a crucial role in the movement of the lower limb joints, affecting both kinetic and kinematic parameters. Since neuromuscular control and coordination heavily rely on proprioceptive information from the body [14], enhancing these aspects in individuals with PFPS can improve their performance and speed up the rehabilitation process.

In this context, time to stabilization (TTS) refers to the person's ability to regain balance when transitioning from dynamic to static positions based on support [15]. It is an element of lower limb motor control, influenced by feed-back from proprioceptive receptors, preprogrammed muscle activation patterns, and reflex muscle responses [16]. This index further contains the complex and coordinated efforts between the sensory and motor systems of the body, as well as the strong contraction chain of the muscles and auxiliary stabilizers in the lower limbs [17]. Therefore, examining the TTS provides much more information about the ability to reach postural control in the event of sudden changes from dynamic to static positions.

Various effective measures have been suggested to improve PFPS, including exercise therapy, muscular strengthening exercises, muscle stretching, electrotherapy, knee braces, and patellar taping [18]. Among them, taping comes in different types and is commonly used in various sports as well as clinical conditions to prevent and treat injuries, as well as to provide support to joints during movement [19, 20]. One of the most distinguished types of such tapes is the Kinesio tape (KT), produced by Kenzo Kase in 1973. This tape can be longitudinally extended up to 140% of its original length. It also creates fewer mechanical and motion restrictions than conventional ones, corrects joint dysfunction and supports muscles [21]. On the other hand, dynamic tape (DT) is being utilized to help patients recover. Compared to other tapes, the DT is a newly developed one with higher elasticity and multidirectional stretch to facilitate sensory input information and enhance proprioception, which, in previous research, was mentioned as its advantage [22]. Due to its elastic nature (tensile load capacity above 200%) and multidirectional stretch (lengthening in all directions) [18], DT as a biomechanical tape has been identified with some properties, such as speed reduction, load absorption, and effectiveness in movements in the original state [22]. It can also prevent many injuries by creating stable positions against high loads and impacts on the PFJ during activities of daily living.

In the research literature review, Kurt et al.'s study found that short-term use of KT was associated with improvements in joint position sense, pain, Kinesio-phobia, and daily limitations [1]. Additionally, Mastamand et al.'s study on the effect of KT on pain and balance in PFPS patients demonstrated that KT is an effective intervention for reducing pain and increasing balance in PFPS patients [23]. On the other hand, the results of Alahmari et al. showed that in people with chronic low back pain, the use of DT controls the processes that lead to back muscle fatigue [21]. Ha et al. (2020) found that applying DT reduced the load and impact on the PFJ during sports activities for participants with PFPS [24]. As DT is a relatively new form of intervention, it has been the subject of fewer research studies. This type of taping shows potential for helping patients with patellofemoral issues due to its associated benefits. Accordingly, this study compares the immediate effects of this taping method with Kinesio tape. If the results show positive effects, researchers can then assess its long-term impact in future studies. Therefore, this study investigates the immediate effects of KT and DT on post-landing task oscillations of the center of pressure (CoP) and time to stabilization (TTS). Our research hypothesis focuses on determining which tape would be more effective for individuals with PFPS based on these variables. The study's results could help choose the most effective taping method for individuals with PFPS.

Materials and Methods

In this randomized crossover trial, 39 women with PFPS, aged 20-30 years were recruited voluntarily after meeting the inclusion criteria. Concerning the G*Power software, version 3.1 outputs, the sample size was then estimated to be 38 individuals, considering the effect size of 0.25, the test power of 0.80 and the significance level of 0.05 [25].

The study's inclusion criteria were as follows: Female gender, age between 20 and 30 years, experiencing knee pain while walking up and down stairs, kneeling and sitting for extended periods with bent knees, and having PFPS in the dominant leg. Additionally, the participants should have had no history of lower limb surgeries, traumas, or neurological, muscular and skeletal diseases, and they should have tested positive for the patellar apprehension test and Clarke test [1, 2, 26]. Moreover, if a participant did not attend the evaluation sessions, they were excluded from the study.

Before the evaluations, the participants received complete explanations about the research procedure and signed an informed consent form for voluntary participation in the study. All stages were conducted according to Helsinki's ethical considerations, The participants entered the study voluntarily, and data was strictly kept secret. Moreover, the participants were informed that they could withdraw from the research at any time without needing to explain.

Study procedure and randomization

The session began by measuring the subjects' height, weight, and body mass index using a BSM370 stadiometer (InBody USA). The participants were evaluated over three separate days. At the beginning of each session, the subjects were randomly assigned to one of three conditions: Without tape (WT or control), KT (ARES Kinesio Tape, South Korea) and DT (United States). Randomization was done through a lottery at the start of each session by an individual not involved in the study, and the participants were unaware of this process. The evaluations included a single-leg stance test on the affected limb and walking down 30-cm stairs using a foot pressure measurement system (FDM-S, Zebris Medical GmbH, Germany). Each participant was evaluated three times under each condition. The maximum and minimum intervals between different test days were two and four, respectively [27].

Taping

In the present study, two tapes of 10 and 15 cm were applied for taping purposes. At first, a 10-cm tape was affixed to the medial condyle of the femur, and the other end was attached to the lateral one from the apex of the patella. This tape was utilized to correct the patellar malalignment. Then, a 15-cm tape was connected to the midpoint of the medial and lateral condyle of the tibia. One end was pulled up upon passing the end of the medial broad muscle, and the second end was pulled up after passing the outer edge of the patella and the outer broad muscle. It was also stretched to cover the knee circumference (Figure 1). During this taping process, the knee had a 30-to-45-degree flexion angle [28, 29].

Center of pressure oscillation recording

To assess the oscillations of the CoP in a static state, the participants stood barefoot on one leg (the taped leg) on the foot pressure measurement system plane. They were then instructed to place their hands on the iliac crest, bend the other leg at the knee and keep it raised, and maintain their single-leg stance for 20 s while looking forward at a visual target located 2 m away [30]. The data related to the oscillations of the CoP in the anteroposterior (AP) and mediolateral (ML) directions, path length (PL), average velocity (AV) and CoP sway area (SA) during the test were also collected. Each participant underwent three tests with a 10-s rest between each repetition. The average value of the results from the three tests was then recorded for each individual. The test was performed under two conditions: With eyes open (EsO) and with eyes closed (EsC).

Time to stabilization calculation

To compute the TTS, the participants were asked to walk down a 30-cm stair with their affected limb bare feet without jumping on the foot pressure measurement system and then maintain their position for 10 s. Each participant was tested three times with a 30-s rest between the repeats [31]. The mean value of the three tests was ultimately recorded for each person.

The information about the TTS and CoP was obtained using the foot pressure measurement system (FDM-S, Zebris Medical GmbH, Germany), which has dimensions of 54.34 cm and 2560 sensors with high sensitivity and a sampling rate of 120 Hz. The Win FDM-S software, version 01.02.09, recorded this information in each test.

Data pre-processing

В

The TTS data was processed using MATLAB software, version R2018b, with a fourth-order Butterworth low-pass filter [32]. From the 10 s of information obtained by the foot pressure measurement system; the mean value of the last two seconds was used as the horizontal line. A degree-three polynomial graph was plotted using the ground reaction force information. The point of intersection between the horizontal line and the cubic polynomial graph was considered the TTS. Figure 2 demonstrates the method for calculating the TTS in the single-leg landing movement.



Figure 1. A) An example of patella kinesio taping, B) An example of patella dynamic taping

PHYSICAL TREATMENTS



Figure 2. Example of calculating time to stabilization

PHYSICAL TREATMENTS

Statistical analysis

Considering the normally distributed data, we used the Shapiro-Wilk test (P>0.05) to examine the data. Then, we employed repeated measures analysis of variance to compare the differences between the test conditions (WT, KT and DT). The data was evaluated using the Mauchly sphericity test to check for sphericity, and the Greenhouse-Geisser correction was applied to assess the significance of the original time effects. The Mauchly sphericity test showed no homogeneity of variance (P<0.05). Additionally, the Bonferroni post hoc test was used to make pairwise comparisons for significant main effects. A significance level of 0.05 was used for all analyses.

Results

Table 1 shows the Mean±SD characteristics of the participants. According to Table 2, the results of the repeated measures analysis of variance for comparing various conditions with each other regarding balance with EsO, ML (F=4.544, P=0.014) and AP oscillations (F=4.021, P=0.022), SA (F=5.716, P=0.005), PL (F=3.780, P=0.025), and AV (F=3.882, P=0.025) of oscillations showed a significant difference (Table 2). In addition, the Bonferroni post hoc test outputs for the pairwise comparison between the conditions in the EsO state suggested a significant difference between the WT condition and the DT in the ML sway (P=0.045), AP sway (P=0.032), SA (P=0.014), PL (P=0.045) and AV of oscillations (P=0.037) (Table 3).

In addition, a significant difference was observed in the results for ML oscillations (F=5.317, P=0.007), AP oscillations (F=5.540, P=0.010), SA (F=7.229, P=0.002), PL (F=6.131, P=0.005) and AV (F=7.573, P=0.001) in the EsC test (Table 2). The Bonferroni post hoc test also

revealed a significant difference between the WT condition and the DT (P=0.015) and between the WT condition and KT application (P=0.028) in the ML oscillations. Similarly, there was a significant difference between the WT condition and the DT (P=0.037) and between the WT condition and KT application (P=0.037) in AP oscillations. Additionally, there was a significant difference between the WT condition and the DT (P=0.007) and between the WT condition and the DT (P=0.007) and between the WT condition and the DT (P=0.045) in SA, between the WT condition and the DT (P=0.022) and between the WT condition and KT application (P=0.010) in PL and between the WT condition and the DT (P=0.006) and the WT condition and the KT application (P=0.016) in AV of oscillations (Table 3).

Moreover, the TTS results (F=9.560, P<0.001) showed a significant difference in the pairwise comparison. There was a notable distinction between the WT condition and the DT (P<0.001) and between the DT and the KT (P=0.007). There was no difference between Kt and WT (P=0.963) (Table 3).

Discussion

This study examined the immediate effects of DT and KT on the oscillations of the CoP and TTS in women with PFPS. The results demonstrated that the open-eye DT intervention significantly enhanced the CoP variables compared to the WT condition. No significant difference between the KT and WT interventions was observed when the eyes were open. However, a significant difference was found between the KT and DT interventions with WT when the eyes were closed. In terms of the TTS, the study showed that DT resulted in a decrease in TTS compared to the WT and KT conditions. No significant difference in TTS was observed between the KT and WT conditions.

Table 1. The participants' characteristics

No. of Participants	Mean±SD				
	Age (y)	Weight (Kg)	Height (cm)	BMI (Kg/m²)	
39	27.92±0.82	63.39±1.55	161.99±0.59	25.05±0.70	
CD. Claudend design	M. D. d				

SD: Standard deviation; BMI: Body mass index.

Meanwhile, DT reduced postural oscillations with and without visual information, while KT reduced postural oscillations only when the eyes were closed. We used a specific taping technique to correct the patella's natural tracking, focusing on applying tape to the vastus medialis oblique and vastus lateralis muscles to take advantage of its beneficial properties. The first tape was applied to the lower part of the patella and between the two femoral condyles to maintain its normal alignment.

Research has shown that patellar gliding, tilting, and rotating are important components of movement related to taping. Patellar gliding refers to the linear distance between the midpoint of the patella and the medial/lateral epicondyle of the femur, while patellar tilting represents the movement of the anterior surface of the patella relative to the femoral condyles. Additionally, the rotation of the patella occurs around an AP axis as the movement of the distal tip of the patella [21, 23, 33, 34]. The second tape was applied to prevent patellar gliding

while surrounding the space of the patella. DT's effect on reducing postural fluctuations in open and closed eye conditions is due to its mechanical support for placing the patella in its correct position. DT, a relatively new adjunct method for treating musculoskeletal issues, was developed by American physical therapist Ryan Kendrick in 2009. It is made of viscoelastic nylon and lycra and can be stretched in four directions. This tape also has strong tensile strength and recoil without a rigid endpoint, which has increased the effectiveness of elastic tapes and their popularity [21].

The study revealed that using KT with open eyes did not impact postural fluctuation, but with closed eyes, the fluctuations decreased. The primary function of KT is its neurophysiological effects. Probably, in the absence of visual information, KT has been able to improve CoP fluctuations and compensate for the lack of vision by increasing the sent afferent information such as joint positional sensory information, stimulat-

Table 2. The results of the repeated measures analysis of variance

Variables		Mean±SD		_		?	
		WT	DT	кт	F	۲	η-
EsO	ML (mm)	10.03±2.88	8.74±1.83	9.67±2.24	4.544	0.014*	0.107
	AP (mm)	21.78±7.39	18.26±4.87	19.90±6.60	4.021	0.022*	0.096
	SA (mm×mm)	186.48±71.90	145.18±46.93	157.94±52.84	5.716	0.005*	0.131
	PL (mm)	232.26±89.78	195.59±52.26	214.88±68.92	3.780	0.025*	0.090
	AV (mm×s)	23.89±9.76	19.92±5.32	21.88±7.02	3.882	0.025*	0.093
EsC	ML (mm)	18.92±4.03	16.70±4.48	16.93±4.84	5.317	0.007*	0.123
	AP (mm)	30.56±9.02	25.42±7.87	26.64±5.98	5.540	0.010*	0.127
	SA (mm×mm)	442.21±161.56	325.03±155.77	370.33±157.12	7.229	0.002*	0.160
	PL (mm)	502.95±166.79	416.11±157.86	430.99±153.99	6.131	0.005*	0.139
	AV (mm×s)	51.17±16.97	41.05±13.55	44.13±16.13	7.573	0.001*	0.166
	TTS (s)	0.51±0.23	0.32±0.15	0.46±0.21	9.560	<0.001*	0.201
	AV (mm×s)	51.17±16.97 0.51±0.23	41.05±13.55 0.32±0.15	44.13±16.13 0.46±0.21	7.573 9.560	0.001* <0.001* PHYSICAI	0.166 0.201

Abbreviations: EsO: Eyes open; EsC: Eyes closed; AP: Anteroposterior; ML: Mediolateral; SA: Sway area; PL: Path length; AV: Average velocity; TTS: Time to stability; SD: Standard deviation.

*Significant difference (P<0.05).

Variables	Eye	Condition (i)	Condition (j)	Р	Eye	Condition (i)	Condition (j)	Р
ML (mm)		T	КТ	0.114		DT	КТ	1
	Open	DI	WT	0.045*	Close		WT	0.015*
		КТ	WT	1		КТ	WT	0.028*
AP (mm)		DT	КТ	0.589		DT	КТ	1
	Open	DI	WT	0.032*	Close		WT	0.037*
		КТ	WT	0.342		КТ	WT	0.037*
SA (mm²)		DT	КТ	0.904		DT	KT	0.336
	Open		WT	0.014*	Close		WT	0.007*
		КТ	WT	0.054		КТ	WT	0.045*
PL (mm)		DT	КТ	0.292		DT	KT	1
	Open		WT	0.045*	Close		WT	0.01*
		КТ	WT	0.672		КТ	WT	0.022*
AV (mm/sec)		т	КТ	0.294		DT lose	KT	10.674
	Open	DI	WT	0.037*	Close		WT	0.016*
		КТ	WT	0.625		КТ	WT	0.006*
TTS (s)		т	КТ	0.007*				
	5)	01	WT	<0.001*				
		КТ	WT	0.963				

Table 3. Bonferroni post hoc test results

PHYSICAL TREATMENTS

Abbreviations: EsO: Eyes open; EsC: Eyes closed; AP: Anteroposterior; ML: Mediolateral; SA: Sway area; PL: Path length; AV: Average velocity; TTS: Time to stability.

*Significant difference (P<0.05).

ing the muscle spindle or the Golgi tendon organ, and strengthening the proprioception of soft tissue to the central nervous system [1, 18, 23]. The KT, designed with a thickness similar to human skin, offers advantages such as creating a local stimulus, aligning fascial tissues, and providing sensory stimulation to assist or limit movements [35]. Additionally, the tape, with its waterproof, breathable fabrics, is popular due to its adhesiveness, ease of use, and lack of allergic reactions. It may also increase local blood flow or lymphatic circulation [18]. Dynamic tape had a mechanical effect in keeping the patella in its groove, while KT had more of a neurophysiological effect.

An important factor investigated in this study was the TTS after the tasks of walking down the stairs and maintaining balance for another 10 s. This index refers to the time (in seconds) the ground reaction force needs to stabilize in a quiet static state after changing from the dynamic to the static position. The ability to stabilize as quickly as possible is generally a positive characteristic because damage reduction is influenced by the effective maintenance of the perceived stability of the body following the change of position [31].

Our study results indicated a significant decrease in TTS after DT compared to the WT and KT conditions. The reduction in the DT was significantly more than the KT, and the DT led to faster TTS. One of the main reasons for these results probably was the better control of the movement of walking down the stairs due to lowering the pain felt by the participants in DT condition. Pain may thus reflexively inhibit the muscles around the knee. Since these muscles are responsible for timely and effective movement responses in controlling the body posture, the effect of DT on improving the ability to maintain balance and reach the TTS faster in people with PFPS could be thus compared to its impacts on balance and pain reduction [23]. Aminaka and Gribbel (2005) found that taping reduced nerve inhibition in the quadriceps by moderating the activity of large sensory nerve fibers, resulting in decreased knee pain [34]. The patella plays a crucial role in enhancing the functions and mechanical benefits of the quadriceps at the front of the femur. Increasing the contact area between the patella and the femur significantly helps distribute compressive forces within the joint. Additionally, the patella provides support for maintaining both static and dynamic stability [9]. Therefore, reducing the pain in this joint by decreasing the contact surface between the patella and the femur and enhancing its proprioception is a key aspect of the knee joint's dynamic stability and motion control system. This could explain the impact of DT on dynamic postural control and reducing the TTS in participants with PFPS in this study [23]. The greater impact of the DT could be further attributed to the materials and their strength, as well as some psychological factors. Of note, the primary mode of action of the DT was mechanical, specifically the deceleration of eccentric work, load absorption, and assistance in movements, while its second mode was neurophysiological [21]. The DT had a more significant effect on reducing the TTS while walking down the stairs. This task required the concentric contraction of the quadriceps and load absorption on the PFJ. The study participants showed better performance in DT conditions. In previous studies, pain perception and lower limb function have been associated with psychological factors and Kinesio-phobia (fear of movement) in patients with PFPS. Thus, Kinesio-phobia in such cases was more closely related to faulty movement patterns and self-reported pain and disability than to PFJ strength and loading [36]. In this regard, Jaffri and Baellow (2023) recommended the use of braces and tapes as a supplementary treatment to alleviate Kinesio-phobia in individuals with PFPS [36]. Therefore, using patellar taping could psychologically help the participants in performing the tasks in the current study. The study results align with Ha et al. (2020) findings, which compared the differences in muscle activity ratio, ground reaction force, and knee valgus angle during single-leg squat and landing due to the use of DT. Ha concluded that the application of DT reduced the load and impact on the PFJ during sports activities for participants with PFPS [24].

Conclusion

The study found that DT reduced CoP variables in EsO and EsC, but KT reduced them only with EsC. Moreover, DT led to a faster TTS than WT and KT conditions. The results indicate that DT application may be more effective in improving balance and postural control in PFPS patients. It is recommended that future research investigate the long-term impact of utilizing this intervention, considering its positive effects.

Study limitations

This randomized crossover trial had some limitations. While previous research had used the landing task to calculate the TTS, this study used the task of walking down the stairs to accommodate the participants' conditions and the need to control the joint load. The study also looked into the immediate effects of the interventions, as DT use had been less examined in previous research, but its positive effects were observed in this study. We suggest investigating the long-term effects of this intervention on patients with PFPS in future studies. It is also recommended that the impact of the DT on other tasks be explored.

Ethical Considerations

Compliance with ethical guidelines

This study was further approved by the Research Ethics Committee of Bu-Ali Sina University, Hamedan, Iran (Code: IR.BASU.REC.1402.013) and archived in the Iranian Registry of Clinical Trials, Tehran, Iran (Code: IRCT20200204046368N10).

Funding

This research did not receive any specific grant from funding agencies in the public, commercial or not for profit sectors.

Authors' contributions

All authors contributed equally to the conception and design of the study, data collection and analysis, interpretation of the results, and drafting of the manuscript. Each author approved the final version of the manuscript for submission.

Conflict of interest

The authors declared no conflict of interest.

Acknowledgments

The authors thanks all patients for participating in this study.

References

- [1] Kurt EE, Büyükturan Ö, Erdem HR, Tuncay F, Sezgin H. Short-term effects of kinesio tape on joint position sense, isokinetic measurements, and clinical parameters in patellofemoral pain syndrome. Journal of physical therapy science Journal of Physical Therapy Science. 2016; 28(7):2034-40. [DOI:10.1589/jpts.28.2034] [PMID] [PMCID]
- [2] Yalfani A, Raisi Z. Comparison of two methods quadriceps muscle strengthening on land and in the water on pain, function, static and dynamic balance in females with patellofemoral pain syndrome. Studies in Sport Medicine. 2013; 5(13):91-108. [Link]
- [3] de Moura Campos Carvalho-E-Silva AP, Peixoto Leão Almeida G, Oliveira Magalhães M, Renovato França FJ, Vidal Ramos LA, Comachio J, et al. Dynamic postural stability and muscle strength in patellofemoral pain: Is there a correlation? Knee. 2016; 23(4):616-21. [DOI:10.1016/j. knee.2016.04.013] [PMID]
- [4] Parijat P, Lockhart TE. Effects of quadriceps fatigue on the biomechanics of gait and slip propensity. Gait & Posture. 2008; 28(4):568-73. [DOI:10.1016/j.gaitpost.2008.04.001]
 [PMID] [PMCID]
- [5] Salavati M, Moghadam M, Ebrahimi I, Arab AM. Changes in postural stability with fatigue of lower extremity frontal and sagittal plane movers. Gait & Posture. 2007; 26(2):214-8. [DOI:10.1016/j.gaitpost.2006.09.001] [PMID]
- [6] Reimer RC 3rd, Wikstrom EA. Functional fatigue of the hip and ankle musculature cause similar alterations in single leg stance postural control. Journal of Science and Medicine in Sport. 2010; 13(1):161-6. [DOI:10.1016/j.jsams.2009.01.001] [PMID]
- [7] Hassan BS, Mockett S, Doherty M. Static postural sway, proprioception, and maximal voluntary quadriceps contraction in patients with knee osteoarthritis and normal control subjects. Annals of the Rheumatic Diseases. 2001; 60(6):612-8. [DOI:10.1136/ard.60.6.612] [PMID] [PMCID]
- [8] Hassan BS, Doherty SA, Mockett S, Doherty M. Effect of pain reduction on postural sway, proprioception, and quadriceps strength in subjects with knee osteoarthritis. Annals of the Rheumatic Diseases 2002; 61(5):422-8. [DOI:10.1136/ ard.61.5.422] [PMID] [PMCID]
- [9] Callaghan MJ, Selfe J, Bagley PJ, Oldham JA. The Effects of Patellar Taping on Knee Joint Proprioception. Journal of Athletic Training. 2002; 37(1):19-24. [PMID]
- [10] Mickel TJ, Bottoni CR, Tsuji G, Chang K, Baum L, Tokushige KA. Prophylactic bracing versus taping for the prevention of ankle sprains in high school athletes: A prospective, randomized trial. The Journal of Foot and Ankle Surgery. 2006; 45(6):360-5. [DOI:10.1053/j.jfas.2006.09.005] [PMID]

- [11] Hewitt BA, Refshauge KM, Kilbreath SL. Kinesthesia at the knee: The effect of osteoarthritis and bandage application. Arthritis and Rheumatism. 2002; 47(5):479-83. [DOI:10.1002/ art.10663] [PMID]
- [12] Akseki D, Akkaya G, Erduran M, Pinar H. [Proprioception of the knee joint in patellofemoral pain syndrome (Turkish)]. Acta Orthopaedica et Traumatologica Turcica. 2008; 42(5):316-21. [DOI:10.3944/AOTT.2008.316] [PMID]
- [13] Grenholm A, Stensdotter AK, Häger-Ross C. Kinematic analyses during stair descent in young women with patellofemoral pain.Clinical Biomechanics. 2009; 24(1):88-94. [DOI:10.1016/j.clinbiomech.2008.09.004] [PMID]
- [14] Bennell KL, Hinman RS, Metcalf BR, Crossley KM, Buchbinder R, Smith M, et al. Relationship of knee joint proprioception to pain and disability in individuals with knee osteoarthritis. Journal of Orthopaedic Research. 2003; 21(5):792-7. [DOI:10.1016/S0736-0266(03)00054-8] [PMID]
- [15] Liu K, Heise GD. The effect of jump-landing directions on dynamic stability. Journal of Applied Biomechanics. 2013; 29(5):634-8. [DOI:10.1123/jab.29.5.634] [PMID]
- [16] Wikstrom EA, Tillman MD, Smith AN, Borsa PA. A new force-plate technology measure of dynamic postural stability: The dynamic postural stability index. Journal of Athletic Training. 2005; 40(4):305-9.
- [17] Linens SW, Ross SE, Arnold BL, Gayle R, Pidcoe P. Postural-stability tests that identify individuals with chronic ankle instability. Journal of Athletic Training. 2014; 49(1):15-23. [DOI:10.4085/1062-6050-48.6.09] [PMID] [PMCID]
- [18] Chang WD, Chen FC, Lee CL, Lin HY, Lai PT. Effects of kinesio taping versus mcconnell taping for patellofemoral pain syndrome: A systematic review and meta-analysis. Evidence-Based Complementary and Alternative Medicine. 2015; 2015:471208. [DOI:10.1155/2015/471208] [PMID] [PM-CID]
- [19] Williams S, Whatman C, Hume PA, Sheerin K. Kinesio taping in treatment and prevention of sports injuries: A meta-analysis of the evidence for its effectiveness. Sports Medicine. 2012; 42(2):153-64. [DOI:10.2165/11594960-000000000-00000] [PMID]
- [20] Christou EA. Patellar taping increases vastus medialis oblique activity in the presence of patellofemoral pain. Journal of Electromyography and Kinesiology. 2004; 14(4):495-504. [DOI:10.1016/S1050-6411(03)00154-8] [PMID]
- [21] Alahmari KA, Rengaramanujam K, Reddy RS, Samuel PS, Tedla JS, Kakaraparthi VN, et al. The immediate and shortterm effects of dynamic taping on pain, endurance, disability, mobility and kinesiophobia in individuals with chronic non-specific low back pain: A randomized controlled trial. Plos One. 2020; 15(9):e0239505. [DOI:10.1371/journal. pone.0239505] [PMID] [PMCID]
- [22] McNeill W, Pedersen C. Dynamic tape. Is it all about controlling load? Journal of Bodywork and Movement Therapies. 2016; 20(1):179-88. [DOI:10.1016/j.jbmt.2015.12.009] [PMID]
- [23] Mostamand J, Shafizadegan Z, Tarrahi MJ, Hosseini ZS. [Comparing the effect of kinesiology taping on dynamic balance and pain of men and women with unilateral patellofemoral pain syndrome (Persian)]. Archives of Rehabilitation. 2021; 21(4):470-87. [DOI:10.32598/RJ.21.4.3071.1]

- [24] Ha TW, Park SH, Lee MM. Comparison of difference in muscle activity ratio, ground reaction force and knee valgus angle during single leg squat and landing according to dynamic taping. Physical Therapy Rehabilitation Science. 2020; 9(4):281-6. [DOI:10.14474/ptrs.2020.9.4.281]
- [25] Yalfani A, Ahmadi M, Gandomi F. The effects of 12-weeks of sensorimotor exercise on pain, strength, pelvic drop, and dynamic knee valgus in males with patellofemoral pain syndrome. Physical Treatments-Specific Physical Therapy Journal. 2020; 10(3):159-68. [DOI:10.32598/ptj.10.3.442.1]
- [26] Piva SR, Fitzgerald K, Irrgang JJ, Jones S, Hando BR, Browder DA, et al. Reliability of measures of impairments associated with patellofemoral pain syndrome. BMC Musculoskeletal Disorders. 2006; 7:33. [DOI:10.1186/1471-2474-7-33] [PMID] [PMCID]
- [27] Yalfani A, Raeisi Z. Prophylactic ankle supports effects on time to stabilization, perceived stability and ground reaction force during lateral landing in female collegiate athletes with chronic ankle instability. BMC Sports Science, Medicine & Rehabilitation. 2021; 13(1):62. [DOI:10.1186/s13102-021-00291-3] [PMID] [PMCID]
- [28] Rehman N, Kashif M, Sajjad AG, Hassan HMJ, Iram H. Immediate effects of kinesio taping on pain in athletes with patellofemoral pain syndrome. Physikalische Medizin Rehabilitationsmedizin Kurortmedizin. 2020; 30(6):344-9. [DOI:10.1055/a-1147-9124]
- [29] Arrebola LS, Teixeira de Carvalho R, Lam Wun PY, Rizzi de Oliveira P, Firmo Dos Santos J, et al. Investigation of different application techniques for Kinesio Taping[®] with an accompanying exercise protocol for improvement of pain and functionality in patients with patellofemoral pain syndrome: A pilot study. Journal of Bodywork and Movement Therapies. 2020; 24(1):47-55. [DOI:10.1016/j.jbmt.2019.05.022] [PMID]
- [30] Yalfani A, Sharifi M, Raeisi Z. [A comparison between two methods of exercise in water and land to improve pain, function, static and dynamic balance in patients with chronic ankle sprain (Persian)]. Sport Sciences and Health Research. 2015; 7(2):175-91. [DOI:10.22059/jsmed.2015.56539]
- [31] Yalfani A, Raeisi Z. Bilateral symmetry of vertical time to stabilization in postural sway after double-leg landing in elite athletes with unilateral chronic ankle sprain. Journal of Foot and Ankle Research. 2022; 15(1):43. [DOI:10.1186/ s13047-022-00552-5] [PMID] [PMCID]
- [32] Wright CJ, Arnold BL, Ross SE. Altered kinematics and time to stabilization during drop-jump landings in individuals with or without functional ankle instability. Journal of Athletic Training. 2016; 51(1):5-15. [DOI:10.4085/1062-6050-51.2.10] [PMID] [PMCID]
- [33] Ghourbanpour A, Talebi GA, Hosseinzadeh S, Janmohammadi N, Taghipour M. Effects of patellar taping on knee pain, functional disability, and patellar alignments in patients with patellofemoral pain syndrome: A randomized clinical trial. Journal of Bodywork and Movement Therapies. 2018; 22(2):493-7. [DOI:10.1016/j.jbmt.2017.06.005] [PMID]
- [34] Aminaka N, Gribble PA. A systematic review of the effects of therapeutic taping on patellofemoral pain syndrome. Journal of Athletic Training. 2005; 40(4):341-51. [PMID]

- [35] Kase K, Wallis J, Kase T. Clinical therapeutic application of the kinesio taping method. Horsham: Kinesio Taping Association International; 2013. [Link]
- [36] Jaffri A, Baellow A. Poor Mental Health Indicators in Individuals With Patellofemoral Pain. Journal of Athletic Training. 2023; 58(10):849-54. [DOI:10.4085/1062-6050-0584.22] [PMID] [PMCID]